Author's response to reviews

Title: Pulmonary arteriovenous malformations presenting as difficult-to-control asthma: a case report

Authors:

Marta Navratil MD (marta.navratil@gmail.com)
Vinko Vidjak MD, PhD, Assistant Professor (vinko.vidjak@gmail.com)
Filip Rubić MD (filiprubic@gmail.com)
Damir Erceg MD (erceg@bolnica-srebrnjak.hr)
Mirjana Turkalj MD, PhD (turkalj@bolnica-srebrnjak.hr)

Version: 4 Date: 22 November 2012

Author's response to reviews: see over
Editors

Journal of Medical Case Reports

Zagreb, November 22, 2012

Dear Editors,

Please find enclosed revised manuscript

MS 5181327307655241 Pulmonary arteriovenous malformations presenting as difficult-to-control asthma: a case report

We have carefully read recommendations of the refree and Editorial Board and we have revised the manuscript accordingly. All latest manuscript changes are visible through MS word’s Track changes.

Following is the list of amendments:
Referee 1

Page 7: „left lower lobe“ has been changed to „left upper lobe“.

Figure 2.: The sentence has been changed to: “Three-dimensional CT angiography revealed a 35-mm vascular malformation within the perihilar left upper lobe with a single feeding artery originating from the pulmonary artery and a single draining vein.”

Figure 5. The sentence has been changed to: “Chest X-ray with visible steels coils after embolotherapy of PAVM malformations within the perihilar left upper lobe and small arteriovenous malformation within the apical segment of the right lobe.”

Editorial Board

The manuscript has been edited by a native-English speaker with scientific expertise.

The figures have been removed from the main manuscript.

We are thankful for your cooperation,

Sincerely

Marta Navratil, and co-authors